Author's response to reviews

Title: Self-reported cold sensitivity in normal subjects and in patients with traumatic hand injuries or hand-arm vibration syndrome

Authors:

Ingela K Carlsson (Ingela.k.carlsson@hotmail.com)
Birgitta Rosén (birgitta.rosen@med.lu.se)
Lars B Dahlin (lars.dahlin@med.lu.se)

Version: 5 Date: 6 April 2010

Author's response to reviews:

Dear Robyn Hughes,

The manuscript has been changed to a single column.
The tables are moved to placed after the reference list.
Figures are uploaded as separate files and numbers are deleted. Figure legends are now part of the manuscript.

Yours sincerely

Ingela Carlsson